Author's response to reviews

Title: Overexpression of CDC25B, CDC25C and phospho-CDC25C (Ser216) in vulvar squamous cell carcinomas are associated with malignant features and aggressive cancer phenotypes

Authors:

Zhihui Wang (zhihui.wang@radiumhospitalet.no)
Claes G Trope (c.g.trope@medisin.uio.no)
Vivi Ann Flørenes (vivi.ann.florenes@radiumhospitalet.no)
Zhenhe Suo (zhenhes@medisin.uio.no)
Jahn M Nesland (j.m.nesland@medisin.uio.no)
Ruth Holm (ruth.holm@radiumhospitalet.no)

Version: 4 Date: 19 May 2010

Author's response to reviews: see over
Please find enclosed the revised version of the manuscript ”Overexpression of CDC25B, CDC25C and phospho-CDC25C (Ser216) in vulvar squamous cell carcinomas are associated with malignant features and aggressive cancer phenotypes.” by Zhihui Wang, Claes G Tropè, Viivi Ann Flørenes, Zhenhe Suo, Jahn M Nesland and Ruth Holm.

We have revised the manuscript according to editors’ recommendations.

1. Additional files: We notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.

We have included the tables in manuscript in a tables section following the references.

2. Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Minor revisions have made.

3. Additional files: Please remove additional files 1 and 2.

Additional files 1 and 2 have been removed.

4. References: the reference list should contain up to 30 authors names for each citation. The term et al. should not be used if there are fewer than 30 authors.

The mode of reference list has changed accordingly.

5. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Proof–reading has done.

Yours sincerely